

Citation: Casey GJ, Tinh TT, Tien NT, Hanieh S, Cavalli-Sforza LT, Montresor A, et al. (2017) Sustained effectiveness of weekly iron-folic acid supplementation and regular deworming over 6 years in women in rural Vietnam. PLoS Negl Trop Dis 11(4): e0005446. https://doi.org/10.1371/ journal.pntd.0005446

Editor: Jennifer F. Friedman, Brown University, UNITED STATES

Received: July 21, 2016

Accepted: March 1, 2017

Published: April 13, 2017

Copyright: © 2017 Casey et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: This study involves human participants resident in rural Vietnam. Data will be made available on request through the WEHI ethics committee. The contact persons are Prof Beverley-Ann Biggs (babiggs@unimelb.edu. au) and Prof Alan Cowman (cowman@wehi.edu. au).

Funding: This project was kindly funded by Atlantic Philanthropies Inc. URL: <u>http://www.</u> atlanticphilanthropies.org/. The funders had no role RESEARCH ARTICLE

Sustained effectiveness of weekly iron-folic acid supplementation and regular deworming over 6 years in women in rural Vietnam

Gerard J. Casey¹, Ta T. Tinh², Nong T. Tien², Sarah Hanieh¹, Luca T. Cavalli-Sforza³, Antonio Montresor⁴, Beverley-Ann Biggs^{1,5}*

1 University of Melbourne, Department of Medicine/RMH at the Doherty Institute, Melbourne, Victoria, Australia, 2 National Institute of Malariology, Parasitology and Entomology, Hanoi, Vietnam, 3 Public Health Nutrition Consultant, previously with World Health Organization, Western Pacific Regional Office, Manila, Philippines, 4 World Health Organization, Geneva, Switzerland, 5 Victorian Infectious Diseases Service, The Royal Melbourne Hospital, Melbourne, Victoria, Australia

* babiggs@unimelb.edu.au

Abstract

Background

Weekly iron-folic acid (IFA) supplementation and regular deworming is effective for the prevention of iron deficiency and anaemia in women of child-bearing age. Between 2006 and 2013, a program of weekly IFA and biannual deworming was implemented in Yen Bai province, Vietnam. In this study we aimed to determine the effectiveness of the program in reducing anaemia and the prevalence of hookworm infection after 72 months (six years).

Methods

This prospective cohort study followed up a cohort of 389 women of child-bearing age from baseline until six years after the introduction of the weekly IFA (one tablet containing 200 mg ferrous sulphate, 0.4mg folic acid) and deworming (one 400mg tablet of albendazole given twice yearly) program (May 2006 to 2012). In each of the six surveys (baseline and five follow-up surveys) we measured haemoglobin and ferritin, and the burden of soil transmitted helminth (STH) infections, and in the 72 month survey we also administered a questionnaire to assess adherence and possible impediments to participating in the program.

Results

Two hundred and fifty six (65.8%) of the original 389 women enrolled in the cohort attended the final 72 month survey. Haemoglobin levels were 122 g/L [95% C.I. 120, 124] at baseline and increased to 135g/L [95% C.I. 133, 138] after 72 months. The prevalence of anaemia was 37.8% [95% C.I. 31.0, 44.7] at baseline and reduced to 14.3% [95% C.I. 9.5, 19.1]. Hookworm infection prevalence, 75.9% [95% C.I. 68.1, 83.8] at baseline, reduced to 10.2% [95% C.I. 5.4, 15.0] with no moderate or heavy intensity infections. Seventy-two percent of participants reported still taking at least 75% of the weekly supplements, and 85.0% had taken the most recent deworming treatment.

in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Discussion

Anaemia rates fell significantly during the six-year program, and STH infections were eliminated as a public health risk. Adherence was well maintained but long-term sustainability is challenging in the absence of ongoing external support.

Author summary

Weekly iron-folic acid (IFA) supplementation combined with regular deworming for women of child bearing age is effective in the prevention of iron deficiency and anaemia. Following a baseline survey, a weekly IFA and regular deworming project was implemented in Yen Bai province, Vietnam in 2006, and after 12 months expanded to the entire province. Haematological parameters, soil transmitted helminth (STH) burden and adherence to the program were monitored periodically until 2012. We found anaemia prevalence fell from 37.8% to 14.3% during the six-year period, and haemoglobin levels increased from 122 g/L to 135g/L. STH infections were essentially eliminated as a public health risk. Seventy-two percent of participants continued to take at least 75% of the weekly supplements, and 85.0% took the most recent deworming treatment. These results show that prevention of anaemia in women of child-bearing age with weekly IFA and regular deworming is feasible and effective over a prolonged period. However, long-term sustainability may be a major challenge in some settings in the absence of ongoing external support.

Introduction

Anaemia (haemoglobin < 120g/L) is estimated to affect 29.0% of non-pregnant women worldwide. [1] Iron deficiency anaemia (IDA) is the most common form of anaemia in many resource-poor settings. Nutritional anaemia can also be caused by vitamin deficiencies (A, B2, B6, folate, B12, C, E) and deficiencies of other minerals—such as copper, zinc, and in some cases selenium; and severe protein-energy malnutrition.[2] Intestinal infections causing diarrhoea, malabsorption and blood loss (especially hookworm infection) may cause depleted body iron stores and worsen the risk of other micronutrient deficiencies by interfering with digestion and absorption (e.g., vitamin A).[3]

The consequences of IDA are most evident in women of child-bearing age and children. In pregnancy IDA has been linked to premature delivery, higher maternal morbidity, and infants with low birth weight (LBW), lower iron stores and higher anaemia rates. [4, 5] Iron deficiency anaemia in infants and young children may lead to impaired development [4] with long term health implications.[6, 7]

Weekly iron-folic acid (IFA) supplementation is now recommended by WHO for nonpregnant women of child-bearing age living in areas with a prevalence of anaemia above 20% [8, 9], with the aim of improving a women's iron and folate status prior to conception. The rationale for using intermittent rather than daily supplemental iron relates to the adverse impact of high levels of luminal iron in the gut and the concept that taking iron less frequently may facilitate absorption of iron by mucosal cells and lead to fewer side-effects [10–12]. WHO also recommends that preventive chemotherapy for soil transmitted helminths (STH) be considered in this group in endemic areas where the prevalence is 20% or higher. [13] However, it is not clear that compliance and effectiveness will be maintained over many years. Long-term sustainability is also of concern, especially when program implementation is dependent on external donor funding. Although the cost of weekly IFA supplementation per woman is estimated to be as little as USD 0.76 cents per year, this still equates to \$76,000.00 per 100,000 women per year, a cost few local health administrations can afford. [14]

Between 2006 and 2013, we initiated and supported a program of weekly IFA supplementation, with regular deworming, to a target population of approximately 250,000 women of child-bearing age in Yen Bai Province, Vietnam.[15–17] Our objectives in this study were to document the adherence and effectiveness of weekly IFA supplementation and deworming on rates of hookworm infection, anaemia, and iron deficiency in a cohort of participants followed from baseline to 72 months (6 years) post intervention. We also sought to identify administrative challenges and barriers to ongoing sustainability.

Methods

Sampling and the intervention

The intervention has been described previously.[18] Briefly, a demonstration project of weekly IFA (one 200mg tablet of Ferrous Sulphate—equivalent to 60mg elemental iron, 0.4mg folic acid) supplementation combined with four monthly deworming (Albendazole 400mg) for non-pregnant women of child bearing age was initiated in May 2006 in two districts in Yen Bai province over a period of 12 months. Tran Yen and Yen Binh districts were selected for the intervention, being easy to reach, and with both Kinh and ethnic minority populations. All nonpregnant women of reproductive age between 16 and 45 years were eligible for the intervention. Supplements were supplied free of charge. Village health workers (VHWs) were central to the delivery of the intervention to women in their communities, and worked closely with the Women's Union to mobilize the target population.

A baseline survey was undertaken in November 2005 in a randomly selected cohort of women who were then followed up at 3 and 12 months post introduction of the intervention. [19] Sample selection for the cohort used a stratified multi-stage cluster design, in which 'probability proportional to size' random sampling was used to select primary sampling units (villages) within each district, and secondary sampling units (women) were selected randomly from each village using provincial lists. A sample size of at least 280 was required in the baseline and follow-up surveys to detect an increase in haemoglobin of 5 g/L with a power of 0.9, a type 1 error of .05, and accounting for clustering with an intraclass correlation of 0.2. This number was also sufficient to detect a reduction in hookworm prevalence of 30%. There was no control group as it was considered unethical to actively withhold IFA supplementation and deworming treatment from this group over several years.

Expansion and implementation of a province-wide weekly IFA/ deworming program

After 12 months of the demonstration project, improvements were seen in all measured indices of women's health: haemoglobin, serum ferritin, anaemia (Hb < 120 g/L), iron deficiency (serum ferritin < 15 mcg/L) and STH prevalence and intensity of infection. [15] As a result, the provincial authorities supported the expansion of the intervention to target reproductive-aged women in all districts of the province (some 250,000 women). The community-based program was administered from the Yen Bai Centre for Malaria Control, through the District Centres for Preventive Medicine to Commune Health Stations and VHWs. The expanded program consisted of a weekly IFA tablet (200mg Ferrous Sulfate) and one tablet of Albendazole

(400mg) given twice a year. Eligible women were encouraged to collect their supplements monthly from their VHW, who recorded the woman's name and date of distribution and advised about side-effects and safe storage of supplements. Albendazole was given as observed treatment on locally designated days either at the commune health station or supervised in the village by a commune health worker. National oversight of the program and support for training and educational material development and production was provided by the National Institute of Malariology, Parasitology and Entomology, which also has responsibility for the national hookworm control program. The Provincial Health Department provided salary support for distribution through the health system. WHO donated albendazole tablets but external donor funding was required for the IFA supplements, training and training materials and educational and promotional materials. Thus, the program was not fully supported by the national health system, remaining partly dependent on external financial and administrative support.

Monitoring and evaluation of the expanded program

The same cohort of women who participated in the baseline, three and 12 month surveys during the demonstration project (as described above) were invited to participate in the 30 month, 54 month, and 72 month follow-up surveys if they were still resident in the same villages.

In all surveys participants were asked to provide a venous blood sample and a faecal sample. Haemoglobin was measured by HemoCue 201+ (HemoCue AB, Angelholm, Sweden) and the STH burden was determined using the Kato-Katz technique as described by Ash et.al. [20] Classification of helminth infection intensity (eggs per gram, EPG) was according to WHO cut offs: 1) Hookworm infections: light = 1–1999, moderate = 2000–3999, heavy \geq 4000 2) Ascaris *lumbricoides*: light = 1–4999, moderate = 5000–49999, heavy \geq 50000, and 3) *Trichuris trichiura* light = 1–999, moderate = 1000–9999, heavy \geq 10000 EPG. For previous surveys, serum ferritin was measured using a sandwich immunoenzymatic assay (IEA; Beckman Coulter Access Reagents, Fullerton, CA). Due to a change in technology at the original laboratory and a subsequent change in the laboratory used, serum ferritin for the 72 month survey was analysed using a Cobas immunoturbidometric test kit (Roche Diagnostics GmbH, D-68298 Mannheim). The structured interview administered during the 72 month survey sought information about whether women had taken the last deworming treatment and how regularly they had taken IFA supplements during the previous 10 weeks, and reasons for noncompliance. It also asked about other factors that may influence adherence such as access to, or provision of, supplements and direct or indirect costs related to travel or lost work time. Adherence with the supplementation regime was defined as taking at least 75% of the supplements provided by the VHW.

Statistical analysis

Data of surveys collected before year 6 have been analysed and reported previously. [21] By pooling all available survey data we extended the former semi-cross-sectional, semi longitudinal panel design with data collected as part of the final survey. Data analysis was conducted using Stata/IC 11.2 (College Station, TX). Cross-sectional summary data was obtained via linear (haemoglobin, log transformed ferritin) and logistic (prevalence) regression models that were fitted per survey time point, incorporating clustering at village level (Huber-White Sandwich estimator) to obtain robust standard errors. Ferritin data was right-skewed and therefore log transformed to normalise the data. Arithmetic mean haemoglobin, geometric mean ferritin, and prevalence (anaemia, iron deficiency, IDA, hookworm, ascaris, trichuris, and total STH infection) with 95% confidence intervals are presented. At each survey time point and for every outcome, the analysis sample consisted of the available outcome data of all eligible women. Data from women who became pregnant during the study were not excluded from the analysis sample. Longitudinal repeated measures mixed linear and logistic regression was used to investigate the change from baseline to year 6, while accounting for clustering at individual and village level.

Adherence data on weekly IFA supplementation and deworming treatment was summarized descriptively by survey as percentages with 95% confidence intervals. Differences in changes over time in anaemia prevalence were examined by ethnic group within district. The odds ratio of being iron deficient at each post-implementation time point compared to baseline was explored. At 72 months, prevalence of iron deficiency was compared between subgroups formed by district, ethnic group, and gender. Changes in prevalence over time by severity of hookworm, *ascaris*, and *trichuris* infection are presented graphically.

Ethics

Extensive consultation was undertaken between the project team, communities and community leaders, as well as liaison with village, district and provincial health staff. Village health workers provided participants with information regarding the surveys and written informed consent was documented at the time of enrolment in the surveys. The National Institute of Malariology, Parasitology and Entomology and the Walter and Eliza Hall Institute of Medical Research and Melbourne Health approved of these consent procedures, which were standard NIMPE consent procedures. Potential recruits received a printed plain language statement and oral consent and documented signatures were obtained prior to participation. Participants were informed that they could withdraw from the study at any time and that withdrawal would not affect their routine medical treatment. Participants were all adult women between the ages of sixteen and forty-five; there were no minors included. The project was approved by the Human Research Ethics Committees of the National Institute of Malariology, Parasitology and Entomology (Hanoi, Vietnam), and the Walter and Eliza Hall Institute of Medical Research and Melbourne Health (Melbourne, Australia) and all ethics committees specifically approved the use of participants of reproductive age between sixteen and forty-five years of age.

Results

The timeline and participation rates for this and previous surveys are shown in Fig 1. Of the 389 women originally surveyed in 2005, 256 returned for the survey in 2012. Two hundred and fifty two women provided blood samples, 216 provided stool samples and six women left before completing the interview. One hundred and seventy eight women (72.0% [95% C.I. 63.6%, 80.4%]) were still taking at least 75% of the supplements they received. A further 29 women received the supplements but either only took them sometimes or gave them to someone else. The forty-three remaining women did not receive the supplements. These latter women were concentrated in six communes in Tran Yen district. Deworming treatment was received by 85.0% (95% C.I.79.5, 90.5) of women (212/249) during the last campaign but two of them did not take it. The adherence over time with taking IFA supplements and deworming treatment is shown in Table 1.

The change in mean haemoglobin and ferritin, and the prevalence of anaemia, iron deficiency anaemia and moderate/heavy STH infection over the 72 month period of program implementation are shown in Table 2. By 2012 the overall mean haemoglobin level was 135g/L [95% C.I. 133, 138]. Based on the mixed-effects model this represents an increase from baseline of 13g/L [95% C.I. 11, 15]. The prevalence of anaemia reduced from 38% [95% C.I. 31, 45] to OS NEGLECTED TROPICAL DISEASES Effectiveness of iron-folic acid supplementation and a deworming program in women in rural Vietnam



¹ WRA = Women of reproductive age defined as between 16 - 45 years ²Hb= Haemoglobin, sFe=serum ferritin, sTfR=soluble transferrin receptor, STH=Soil transmitted helminths

³ WIFS = Weekly Iron Folic acid Supplementation

Fig 1. Timeline of surveys and intervention (adapted from [21]). Notes: ¹ WRA = Women of Reproductive Age defined as between 16–45 years, ²Hb = Haemoglobin, sFe = serum Ferritin, sTfR = soluble Transferrin Receptor, STH = STHs, ³ WIFS = Weekly Iron Folic acid Supplementation.

https://doi.org/10.1371/journal.pntd.0005446.g001

	3m survey Adherence		12m : Adhe	survey erence	30m Adhe	survey erence	54m s Adhe	survey erence	72m survey Adherence		
	N	Per cent (95% C.I.)	N	Per cent (95% C.I.)	N	Per cent (95% C.I.)	N	Per cent (95% C.I.)	N	Per cent (95% C.I.)	
Weekly IFA supplementation	243	97.9 (95.8, 100)	204	98.5 (96.8, 100)	305	93.1 (87.3, 98.9)	285	76.1 (67.9, 84.4)	250	72.0 (63.6, 80.4)	
Deworming treatment	246	89.0 (85.1, 92.9)	210	88.6(81.7, 95.5)	306	87.9 (80.1, 95.7)	288	95.5 (92.8, 98.1)	247	85.0 (79.5, 90.5)	

Table 1. Adherence¹ with weekly IFA supplementation and deworming treatment over time².

Notes:

¹ Defined as the percentage of women taking at least 75% of the supplements provided.

² No women were taking supplements at baseline.

https://doi.org/10.1371/journal.pntd.0005446.t001

Table 2. Mean haemoglobin, geometric mean ferritin and prevalence of anaemia, iron deficiency and iron deficiency anaemia, and prevalence of STH infections by WHO categories over time. [32]

	F	Baseline Prevalence	3 r	nonth survey Prevalence	12	month survey Prevalence	30	month survey Prevalence	54	month survey Prevalence	72 month survey Prevalence	
	N	Mean (95% C.I.)	N	Mean (95% C.I.)	N	Mean (95% C.I.)	N	Mean (95% C.I.)	N	Mean (95% C.I.)	N	Mean (95% C.I.)
Haemoglobin (g/ L)	354 ¹	122 (120, 125)	242	126 (124, 128)	207	130 (128, 132)	303	130 (128, 132)	285	131 (128, 134)	252	135 (133, 138)
Ferritin (mcg/L)	334 ¹	28 (23.9, 32.7)	229	37.4 (32.5, 43.0)	200	47.6 (41.9, 54.2)	298	52.4 (45.1, 60.8)	268	53.9 (46.5, 62.4)	252	47.1 (38.8, 57.3)
Anaemia	354	37.8 (31.0, 44.7)	242	26.4 (19.7, 33.2)	207	19.3 (13.2, 25.4)	303	18.8 (13.8, 23.8)	285	17.9 (12.5, 23.3)	252	14.3 (9.5, 19.1)
Iron deficiency	334	25.0 (18.5, 27.6)	229	13.1 (8.7, 17.5)	200	7.0 (3.4, 10.5)	298	9.1 (5.8, 12.3)	268	8.2 (4.9, 11.5)	252	17.1 (12.4, 21.7)
IDA	334	14.4 (9.6, 19.1)	229	6.1 (3.6, 8.7)	200	2.0 (0.1, 3.9)	297	3.7 (1.3, 6.1)	268	3.0 (0.6, 5.4)	252	4.0 (1.5, 6.4)
Hookworm	374 ¹		210		185		187		260		216	
- light infection		59.6(54.6, 64.6)		52.9 (46.1 59.6)		27.6 (21.1, 34.0)		11.8 (7.1, 16.4)		11.5 (7.6, 15.4)		10.1 (6.1, 14.2)
- Mod/Heavy		16.3 (10.1, 12.6)		4.3, (1.5, 7.0)		2.1 (1.1, 4.3)		10.0, (4.0, 16.3)		0		0
Ascaris	374		210		185		187		260		216	
- light infection		12.0 (8.7, 15.3)		5.2 (2.2, 8.3)		3.8 (1.0, 6.5)		2.7 (0.4, 5.0)		4.6 (0.2, 7.2)		1.4 (0.0, 2.9)
- Mod/heavy		7.5, (4.8, 10.2)		1.4, (0, 3.0)		0.5, (0, 1.6)		1.6, (0, 3.4)		0		1.0 (0, 2.2)
Trichuris	374		210		185		187		260		216	
- Light infection		27.3 (22.7, 31.8)		21.9 (16.3, 27.5)		11.4 (6.8, 15.9)		6.4 (2.9, 9.9)		2.7 (0.7, 4.7)		2.3 (0.3, 4.3)
- Mod/Heavy		2.1, (0.7, 3.6)		0		0		3.2, (0.7, 5.8)		0		0
Total STH infection	374	83.7 (79.0, 87.4)	210	67.1 (60.8, 73.5)	185	35.7 (28.7, 42.6)	187	28.3 (21.9, 34.8)	260	16.5, (12.0, 21.1)	216	13.9 (9.3, 18.5)

¹. Of the 389 eligible women, the proportion of women with haemoglobin measurements at all survey time points was 26.5%. The proportion with five, four, three, two, one, or no haemoglobin measurement (s) was 22.9%, 20.0%, 14.7%, 9.5%, 5.9%, and 0.5%, respectively. A similar missing data pattern was observed for ferritin. For hookworm, the proportion of eligible women with data on hookworms at all survey time points was 17.7%. The proportion with five, four, three, two, one, or no data on hookworms was 18.8%, 16.7%, 19.8%, 15.7%, 10.3%, and 1.0% respectively. Pregnant women were included in the analysis.

https://doi.org/10.1371/journal.pntd.0005446.t002





https://doi.org/10.1371/journal.pntd.0005446.g002

14% [95%C.I. 9, 19]. Iron deficiency anaemia reduced from 14% [95%C.I. 10, 19] to 3.9% [95% C.I. 2, 6].

The relative change in anaemia over time by district and ethnic group of Kinh or non-Kinh is shown in Fig 2. The anaemia prevalence dropped significantly from previous levels in all population groups, in the first 12 months of the intervention. In the following years anaemia prevalence continued to decrease significantly, in both the Kinh and non-Kinh ethnic groups, in Yen Binh district. However in Tran Yen district it remained static in the following 5 years, between 11 and 17% for the Kinh, and between 21 and 29% for the non-Kinh. The prevalence of iron deficiency in the 72 month survey was 43/252 (17%, [95%C.I. 10, 24]). This was an increase on previous levels and corresponded to falling adherence rate (Fig 3). There was a higher prevalence of iron deficiency in Tran Yen, 30/140 (21%, [95% C.I. 15, 28]) compared to Yen Binh, 13/112 (12%, [95% C.I. 6, 18]). Iron supplements were reportedly taken by 25/29 (86%, [95% C.I. 73, 100]) of iron deficiency was more prevalent among women from ethnic minority groups (non-Kinh) 23/95 (24%, [95% C.I. 16, 33]) than Kinh women, 20/157 (13%, [95% C.I. 7, 18]). The main reason for not taking supplements was unavailability, but some women also reported not needing them because they felt well.

The overall prevalence of STH infection fell from 83.7% [95% C.I. 77.2, 90.2] to 13.9% [95% C.I. 8.7, 19.1], and hookworm infection from 75.9% [95% C.I. 68.1, 83.8] to 10.2% [95% C.I. 5.4, 15.0], while moderate and heavy infections were essentially eliminated (Fig 4).

Discussion

We report the effectiveness of a community-wide weekly IFA supplementation and regular deworming program in a population of non-pregnant rural Vietnamese women after 72

PLOS NEGLECTED TROPICAL DISEASES Effectiveness of iron-folic acid supplementation and a deworming program in women in rural Vietnam



Fig 3. Odds ratios for iron deficiency compared to baseline and prevalence of adherence to taking iron tablets.

https://doi.org/10.1371/journal.pntd.0005446.g003



Fig 4. Change in Soil Transmitted Helminth (STH) burden during the implementation of the WIFS/ deworming program (2006–2012).

https://doi.org/10.1371/journal.pntd.0005446.g004

months. It is one of the few studies to evaluate an *intermittent* iron supplementation and deworming program for nonpregnant women over a period of many years and offers unique insights into the effectiveness and sustainability of the WHO-recommended approach to prevention of iron-deficiency anaemia in this population. Haemoglobin levels remained well above the baseline mean and the prevalence of anaemia continued to fall in most sections of the population. Moderate and heavy intensity STH infections were virtually eliminated and only 10% of women still had light infections (mostly hookworm).

In 2011, WHO recommended weekly IFA supplementation for non-pregnant women of reproductive age in populations with an anaemia prevalence of 20% or higher, given in three month cycles with a three month gap between each cycle. [8] In the Yen Bai setting, it took 12 months of weekly supplementation for the population prevalence of anaemia to drop below 20% and up to 6 years (72 months) for the prevalence to be significantly below the 20% level. This slow rate of decline in anaemia levels has also been noted in other studies [22, 23] and suggests greater benefits if weekly or biweekly supplementation is given continuously, at least in the first year, and continued for several years.

The increased risk of iron deficiency in women of child bearing age is often compounded by hookworm infection in endemic areas. WHO has previously recommended that this group be included in preventive chemotherapy programs for STH (except during the first trimester of pregnancy), and suggest achieving synergy by packaging this intervention with other interventions. [13] These guidelines are currently being updated and will provide more detail to assist program managers considering this intervention. [24]

We observed reduced but still reasonable adherence with weekly IFA after 54 and 72 months (76% and 72% respectively), suggesting that the program remained popular with the target population. However, despite the relatively high reported adherence rates in Tran Yen district we noted a rise in the prevalence of iron deficiency since the 54 month survey. We found that the supply of IFA supplements had been interrupted in certain communes in Tran Yen district during this time, which may explain the rise in iron deficiency in this district. The failure to achieve a consistent supply of supplements in these communes may be due to inadequate training of new health staff over the 6 years of implementation as there was considerable turnover of commune staff and VHWs (GC personal communication). It is important to note that STH infections of moderate and heavy intensity for any STH species remained at or less than 1%. As these heavier worm burdens infections are main cause of morbidity, we conclude that the deworming intervention virtually eliminated all STH-related morbidity.

Limitations of the study included a reduced participation rate in the later surveys, in spite of the efforts of local village and commune health workers publicising the surveys. This was most likely due to the long follow-up period, and the movement of some families out of the area. The relatively high loss to follow-up may have biased our estimates of adherence and effectiveness, as non-adherent women may have chosen not to take part, while healthier adherent women may have remained engaged. As well, women who were feeling tired, or had illness, and those with poorer economic circumstances may have been less likely to adhere and/or to attend for surveys. This would have resulted in a healthier cohort at the end of the program, making the intervention appear more effective. Other limitations were that adherence data relied on self-reporting rather than objective documentation, and women who attended the survey may have exaggerated their adherence. Ferritin levels were measured using different methodology in this, compared to previous surveys. However both were commercial assays and we have no reason to believe that this change contributed to the lower mean ferritin observed in this survey. We did not include a control group as the research team and provincial authorities felt this would be unethical for a long term program. However, we are unaware of significant improvements in economic conditions during the six year period that may have

accounted for the results presented here. Indeed, the global financial crisis commenced soon after program expansion and so deterioration in community living standards may have been expected. Improvements in main road infrastructure did occur but we did not observe a change in living conditions at village level (Casey, personal communication). The project was conducted in a remote rural region of Vietnam and may not be generalizable to other areas or ethnic groups where the prevalence and causes of anaemia may differ.

The sustainability of weekly IFA/deworming programs for the large populations for whom they are recommended is country-specific. The program in Yen Bai province was mainly externally funded, and so was never fully incorporated as a national or provincially-funded program. In post program debriefings, provincial health and finance officials emphasised that, while the program was well accepted by the population and effective and cheap on a per person basis (0.76USD/woman/year), the cost of supplying weekly supplements to the target population (approx. \$200,000 per annum) was beyond the capacity of the province's health budget. While they were prepared to cover the human resource distribution costs, they were not able to support purchase of IFA supplements, development and production of educational materials, and training (G Casey, T Tinh, personal communication). Multiple micronutrient supplementation is even more expensive, even though it may be indicated in settings with higher rates of micronutrient deficiencies. [25]

There are however encouraging signs for the long-term sustainability of community-based WIFS/deworming programs in some other countries, especially India.[17, 26] Based on sound evidence from the field, [23, 27] the Indian Government has produced an operational framework for universal weekly IFA supplementation for adolescents in school and adolescent females not attending school. [28] Responsibility for the national program, from policy formulation and resource allocation to monitoring and review, has been allocated to the Ministry of Health and Family Welfare [29]. Given that this program is projected to cover 130 million adolescents, it may encourage more countries with at-risk populations to provide resources for similar national programs. Likewise, in Cambodia, the use of weekly iron and folic acid supplementation has been progressively extended, over a 10 year period, to cover most schools (LTC-S personal communication).

A revolving fund approach can help sustain the program, by selling the supplements. This approach was successfully used in the weekly IFA supplementation programme of Hai Duong province, Viet Nam, where the supplements were sold to non-pregnant women through the Women's Union network and provided free of charge when women were pregnant, according to the Vietnamese health policy [30]. Over a year, a non-pregnant woman would spend the equivalent of US\$0.96, which was acceptable for rural women in Vietnam. Funds gained from the sales of the supplements were used to pay for an incentive for Women's Union collaborators to sell the supplement (20%), and for management costs and regular communication and promotion activities in the communes (30%). The remaining 50% was held in a local bank under the supervision of a district steering committee, and used to purchase new supplements, to continue the programme beyond its initial financing period. [30]

A review of weekly IFA supplementation programs conducted in Cambodia, The Philippines and Viet Nam concluded that women are willing and able to purchase supplements when they are widely available and affordable, including in poor rural areas and schools. [31] In the Cambodian factories, where supplements had to be provided free of charge because local laws forbade their sale, WRA asked that the supplements be sold outside the factories so that they could continue taking them in the future. In each country, the programme's success led to expanding weekly iron-folic acid supplementation through larger-scale programmes. The variety of social marketing and community mobilization strategies used in the three countries in schools, factories, and communities (discussed in the above mentioned programme review) provide valuable lessons for replicating this approach in other countries. [31]

An analysis of 10 weekly IFA programmes for the prevention and control of anaemia in women, which took place in 6 countries, confirmed that high compliance in taking the supplements can be achieved, irrespective of supervision, provided recipients are convinced of the benefits through an effective communication strategy, with the participation of several stakeholders, and a system in place for monitoring consumption.[17]

In conclusion, the program of free weekly IFA supplementation and regular deworming for women of reproductive age in Yen Bai province ran successfully for 6 years with external inputs of supplements, training and education. It was well received by the population, with good adherence, and resulted in major reductions in anaemia and STH infection. Sustainability will probably require full integration into Vietnam's national health system. A complementary approach to be considered, successfully used both elsewhere in Viet Nam and in other countries, is to sell the supplements at an affordable price while promoting them through social marketing, thus creating and maintaining demand for the product.

Supporting information

S1 Checklist. Strobe checklist. (DOC)

Acknowledgments

The authors would like to acknowledge the support of Yen Bai People's Committee, Yen Bai Malaria Control Office and the past and present Directors Dr Phu B. Luong and Dr Thu Hoang, Tran Yen and Yen Binh Department of Preventive Medicine staff, commune health station staff and the village health workers and their communities, the women of Tran Yen and Yen Binh who participated in these surveys. In addition we would like to acknowledge Dr Hung M. Nguyen who provided input into the design and implementation plan, and oversaw the NIMPE team that implemented the project in Yen Bai Province, Dr Phuc Q. Tran from NIMPE for his assistance and support, the NIMPE and Yen Bai survey teams for their good humour and persistence through fair weather and foul, Alex Gorelik and Dr Sabine Braat for statistical advice, and Beth Hilton-Thorp and Christalla Hajisava for their invaluable contributions. The authors thank Keith Westbury and scientists at SEALS Australia for their advice and assistance with sample laboratory analysis.

Author Contributions

Conceived and designed the experiments: GJC BAB TTT LTCS AM.

Performed the experiments: GJC TTT NTT.

Analyzed the data: GJC SH.

Contributed reagents/materials/analysis tools: GJC.

Wrote the paper: GJC.

SH: contributed to the creation of figures and tables.

References

- WHO. The global prevalence of anaemia in 2011. Geneva 2015. http://apps.who.int/iris/bitstream/ 10665/177094/1/9789241564960_eng.pdf?ua=1.
- Kraemer K. Nutritional Anaemia. Basel, Switzerland: Sight and Life Press; 2007. <u>http://pdfsr.com/pdf/nutritional-anemia</u>.
- Koski KG, Scott ME. Gastrointestinal nematodes, nutrition and immunity: breaking the negative spiral. Annu Rev Nutr. 2001; 21:297–321. https://doi.org/10.1146/annurev.nutr.21.1.297 PMID: 11375439
- Rahman MM, Abe SK, Rahman MS, Kanda M, Narita S, Bilano V, et al. Maternal anemia and risk of adverse birth and health outcomes in low- and middle-income countries: systematic review and metaanalysis. Am J Clin Nutr. 2016; 103(2):495–504. https://doi.org/10.3945/ajcn.115.107896 PMID: 26739036
- WHO/CDC. Assessing the iron status of populations: Report of a joint World Health Organization/Centers for Disease Control and Prevention technical consultation on the assessment of iron status at the population level. Geneva: WHO/CDC, 2005.
- Darnton-Hill I, Webb P, Harvey PWJ, Hunt JM, Dalmiya N, Chopra M, et al. Micronutrient deficiencies and gender: social and economic costs. Am J Clin Nutr. 2005; 81(5):1198S–205. PMID: <u>15883452</u>
- Gyles CL, Lenoir-Wijnkoop I, Carlberg JG, Senanayake V, Gutierrez-Ibarluzea I, Poley MJ, et al. Health economics and nutrition: a review of published evidence. Nutr Rev. 2012; 70(12):693–708. <u>https://doi.org/10.1111/j.1753-4887.2012.00514.x PMID: 23206283</u>
- 8. WHO. Guideline: Intermittent iron and folic acid supplementation in menstruating women. Geneva: World Health Organization, 2011.
- WHO. Weekly iron-folic acid supplementation (WIFS) in women of reproductive age: its role in promoting optimal maternal and child health. Position statement. Geneva: World Health Organization. 2009.
- Kortman GA, Raffatellu M, Swinkels DW, Tjalsma H. Nutritional iron turned inside out: intestinal stress from a gut microbial perspective. FEMS Microbiol Rev. 2014; 38(6):1202–34. https://doi.org/10.1111/ 1574-6976.12086 PMID: 25205464
- Moretti D, Goede JS, Zeder C, Jiskra M, Chatzinakou V, Tjalsma H, et al. Oral iron supplements increase hepcidin and decrease iron absorption from daily or twice-daily doses in iron-depleted young women. Blood. 2015; 126(17):1981–9. https://doi.org/10.1182/blood-2015-05-642223 PMID: 26289639
- 12. Viteri FE. Iron supplementation for the control of iron deficiency in populations at risk. Nutr Rev. 1997; 55(6):195–209. PMID: 9279056
- 13. WHO. Preventive chemotherapy in human helminthiasis: coordinated use of anthelminthic drugs in control interventions: a manual for health professionals and programme managers. Geneva: 2006.
- Casey GJ, Sartori D, Horton SE, Phuc TQ, Phu LB, Thach DT, et al. Weekly Iron-Folic Acid Supplementation with Regular Deworming Is Cost-Effective in Preventing Anaemia in Women of Reproductive Age in Vietnam. PLoS ONE. 2011; 6(9):e23723. https://doi.org/10.1371/journal.pone.0023723 PMID: 21931611
- Casey G J, Phuc T, MacGregor L, Montresor A, Mihrshahi S, Thach T, et al. A free weekly iron-folic acid supplementation and regular deworming program is associated with improved hemoglobin and iron status indicators in Vietnamese women. BMC Public Health. 2009; 9(1):261.
- Mihrshahi S, Casey GJ, Montresor A, Phuc TQ, Thach DT, Tien NT, et al. The effectiveness of 4 monthly albendazole treatment in the reduction of soil-transmitted helminth infections in women of reproductive age in Viet Nam. Int J Parasitol. 2009; 39(9):1037–43. <u>https://doi.org/10.1016/j.ijpara.</u> 2009.01.013 PMID: 19324046
- WHO Regional Office for the Western Pacific. Weekly iron and folic acid supplementation programmes for women of reproductive age: An analysis of best programme practices Manilla, Phillipines. 155pp 2011. http://www.wpro.who.int/publications/docs/FORwebPDFFullVersionWIFS.pdf.
- Phuc TQ, Mihrshahi S, Casey GJ, Phu LB, Tien NT, Caruana SR, et al. Lessons learned from implementation of a demonstration program to reduce the burden of anemia and hookworm in women in Yen Bai Province, Viet Nam. BMC Public Health. 2009; 9:266. https://doi.org/10.1186/1471-2458-9-266
 PMID: 19638191
- Pasricha SR, Caruana SR, Phuc TQ, Casey GJ, Jolley D, Kingsland S, et al. Anemia, iron deficiency, meat consumption, and hookworm infection in women of reproductive age in northwest Vietnam. Am J Trop Med Hyg. 2008; 78(3):375–81. Epub 2008/03/14. PMID: 18337329
- Ash LR, Orihel TC, Savioli L, Sin MA, Montresor A, Renganathan E. Training manual on diagnosis of intestinal parasites—tutor's guide Geneva: World Health Organization; 2004 [cited 2007 2 August]. http://www.who.int/publications/en/.

- 21. Casey GJ, Montresor A, Cavalli-Sforza LT, Thu H, Phu LB, Tinh TT, et al. Elimination of Iron Deficiency Anemia and Soil Transmitted Helminth Infection: Evidence from a Fifty four Month Iron-Folic Acid and De-worming Program. PLoS Negl Trop Dis. 2013; 7(4).
- Bansal PG, Toteja GS, Bhatia N, Vikram NK, Siddhu A. Impact of weekly iron folic acid supplementation with and without vitamin B12 on anaemic adolescent girls: a randomised clinical trial. Eur J Clin Nutr. 2016; 70(6):730–7. https://doi.org/10.1038/ejcn.2015.215 PMID: 26695724
- Vir SC, Singh N, Nigam AK, Jain R. Weekly iron and folic acid supplementation with counseling reduces anemia in adolescent girls: A large-scale effectiveness study in Uttar Pradesh, India. Food Nutr Bull. 2008; 29(3):186–94. https://doi.org/10.1177/156482650802900304 PMID: 18947031
- 24. WHO. WHO guideline development group meeting-the control of soil-transmitted helminths in high-risk groups. Preventive chemotherapy in preschool children, school-age children and women of childbearing age. 2016. http://www.who.int/nutrition/events/2016_meeting_guidelinedevelopmentgroup_ deworming_13to15april/en/).
- 25. Ahmed F, Khan MR, Akhtaruzzaman M, Karim R, Williams G, Banu CP, et al. Effect of long-term intermittent supplementation with multiple micronutrients compared with iron-and-folic acid supplementation on Hb and micronutrient status of non-anaemic adolescent schoolgirls in rural Bangladesh. Br J Nutr. 2012; 108(8):1484–93. https://doi.org/10.1017/S0007114511006908 PMID: 22244349
- 26. Evaluation of the Weekly Iron and Folic Acid Supplementation (WIFS) Program—(2004–2014) Bhutan. [Internet]. UNICEF. 2014. http://www.unicef.org/evaldatabase/index_80824.html.
- Dwivedi A, Schultink W. Reducing anaemia among Indian adolescent girls through once weekly supplementation with iron and folic acid. Lavenham Press, United Kingdom: 2006.
- Ministry of Health and Family Welfare. Operational Framework: Weekly Iron and Folic Acid Supplementation Programme for Adolescents. New Delhi: Government of India; 2012.
- National Health Mission. The Weekly Iron and Folic Acid Supplementation (WIFS) Programme Government of India; 2015. http://nrhm.gov.in/nrhmcomponnets/reproductive-child-health/adolescent-health/ wifs.html.
- 30. Khan NC, Thanh HT, Berger J, Hoa PT, Quang ND, Smitasiri S, et al. Community mobilization and social marketing to promote weekly iron-folic acid supplementation: a new approach toward controlling anemia among women of reproductive age in Vietnam. Nutr Rev. 2005; 63(12 Pt 2):S87–94. PMID: 16466084
- Cavalli-Sforza T, Berger J, Smitasiri S, Viteri F. Weekly iron-folic acid supplementation of women of reproductive age: impact overview, lessons learned, expansion plans, and contributions toward achievement of the millennium development goals. Nutr Rev. 2005; 63(12 Pt 2):S152–8. PMID: 16466092
- WHO. Monitoring helminth control programmes. Geneva: World Health Organization. WHO/CDS/CPC/ SIP/99.3, 1999.

University Library



A gateway to Melbourne's research publications

Minerva Access is the Institutional Repository of The University of Melbourne

Author/s:

Casey, GJ; Tinh, TT; Tien, NT; Hanieh, S; Cavalli-Sforza, LT; Montresor, A; Biggs, B-A

Title:

Sustained effectiveness of weekly iron-folic acid supplementation and regular deworming over 6 years in women in rural Vietnam

Date:

2017-04-01

Citation:

Casey, G. J., Tinh, T. T., Tien, N. T., Hanieh, S., Cavalli-Sforza, L. T., Montresor, A. & Biggs, B. -A. (2017). Sustained effectiveness of weekly iron-folic acid supplementation and regular deworming over 6 years in women in rural Vietnam. PLOS NEGLECTED TROPICAL DISEASES, 11 (4), https://doi.org/10.1371/journal.pntd.0005446.

Persistent Link:

http://hdl.handle.net/11343/192456

File Description: Published version License: CC BY